Author's response to reviews

Title: Exploring patterns of melanoma recurrence in Northeast Scotland to inform the introduction a digital self-examination intervention

Authors:

Rhona L Auckland (blue-eyedrolo@hotmail.com)
Patrick R Wassell (p.wassell.08@aberdeen.ac.uk)
Susan Hall (s.hall@abdn.ac.uk)
Marianne C Nicolson (mnicolson@abdn.ac.uk)
Peter Murchie (p.murchie@abdn.ac.uk)

Version: 6 Date: 27 February 2014

Author's response to reviews:

Dear Editor

I attach yet another revised manuscript responding to two formatting queries on our accepted manuscript.

Please note the study approval section has been amended and the associated study approval confirmation form is being forwarded with the revised manuscript.

Sincerely

Peter